

RESEARCH ARTICLE/ÉTUDE ORIGINALE

A Thaw in the Chill? Charities and Public Policy Engagement in Canada¹

John D. Cameron¹ (D) and Lauchlan T. Munro² (D)

¹Department of International Development Studies, McCain Building, Room 3028, 6135 University University Ave., Dalhousie University, Halifax, Nova Scotia B3H 4R2, Canada and ²School of International Development and Global Studies, Social Sciences Building, 120 University Private, Room 8005, University of Ottawa, Ottawa, Ontario K1N 6N5, Canada

Corresponding author: John Cameron; Email: John.Cameron@dal.ca

Abstract

This article examines the effects of changes in 2018–19 to the Income Tax Act and Canada Revenue Agency (CRA) regulations that were ostensibly intended to facilitate public policy engagement by Canadian charities. The article examines a case study of charities in the international development sector through interviews with charity leaders and quantitative analysis of data from the Canada Revenue Agency, Office of the Commissioner of Lobbying, and House of Commons Standing Committees. The article finds that the 2019 change in CRA regulations had very little effect on policy engagement by international development charities. Rather, a series of other factors continue to shape and constrain policy engagement by charities—including concerns about the future repoliticization of the CRA, misunderstandings of the regulations, difficulties fundraising for public policy work, fears of jeopardizing federal government funding, and a strategic preference for insider approaches to policy advocacy.

Résumé

En 2018–19, le Parlement canadien et l'Agence du revenu du Canada (ARC) ont adopté un nouveau cadre juridique visant à faciliter l'engagement des organismes de bienfaisance canadiens dans les débats concernant les politiques publiques. Cet article a comme objectif d'enquêter les effets de ce changement. L'article est une étude de cas des organismes de bienfaisance du secteur du développement international au moyen d'entretiens qualitatifs avec des responsables d'organismes de bienfaisance et d'une analyse quantitative des données de l'ARC, du Commissariat au lobbying et des comités parlementaires permanents. Les modifications apportées en 2018-19 au cadre juridique ont eu très peu d'impact sur l'engagement politique des organismes de bienfaisance œuvrant dans le domaine du développement international. Au contraire, une série d'autres facteurs continuent de façonner et de limiter l'engagement politique des organismes de bienfaisance, notamment les inquiétudes concernant la politisation future de l'ARC, les malentendus concernant le contenu des règlements, les difficultés de collecte de fonds pour financer

© The Author(s), 2025. Published by Cambridge University Press on behalf of the Canadian Political Science Association (l'Association canadienne de science politique) and/et la Société québécoise de science politique. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.

l'engagement dans la politique publique, les craintes de mettre en péril le financement fédéral et une préférence stratégique pour le plaidoyer en dehors de la sphère publique.

Key words: charities; public policy; advocacy; Canada Revenue Agency

Mots-clés: organismes de bienfaisance; politique publique; plaidoyer; Agence du revenu du Canada

In December 2018, after decades of debate, the Canadian Parliament amended the Income Tax Act (ITA) to eliminate restrictions on the "political activities" of registered charities. Then, in January 2019, the Canada Revenue Agency (CRA) issued new guidance on 'Public Policy Dialogue and Development Activities (PPDDAs)" by Canadian charities. The new CRA guidance states that:

A charity may engage in unlimited PPDDAs that further its stated charitable purpose(s), provided the charity never directly or indirectly supports or opposes a political party or candidate for public office... In other words, a charity is free to advocate for retaining, opposing, or changing any law, policy, or decision of government in furtherance of its stated charitable purpose (CRA, 2019).²

From a legal perspective, the changes to the ITA and the CRA regulations mark a significant shift in state-civil society relations by allowing charities to freely engage in activities to influence public policy making, so long as those activities are non-partisan and related to the charity's stated purpose. Until 2019, Canadian law had restricted charities from allocating more than 10 per cent of their annual revenue to what the CRA called "political activities"—involving any calls for public action with the goal of influencing government laws, regulations or policies. Many charity sector leaders had claimed that the old CRA regulations, and especially their heightened enforcement from 2012 to 2015 under the Conservative government, had created a "chill effect" on policy engagement (Beeby, 2014; CCIC, 2016a; Consultation Panel on the Political Activities of Charities, 2017; Lasby and Cordeaux, 2016: 16–18). Implicit in these claims was the promise that more charities would engage more seriously in policy debates if the rules were relaxed.

However, five years after the legal changes, policy engagement by Canadian charities does not appear to have increased. Our data suggest that very few charities engaged seriously and consistently in federal policy processes before or after the 2018–2019 legal changes. Since the changes in law could reasonably be expected to result, at least eventually, in increased policy engagement, we have an intellectual puzzle: Why have Canadian charities not become more engaged with federal policy making, despite the new legal opportunities to do so?

Solving this puzzle, at least for charities in the international development sector, is one of the contributions of this article. We investigate this puzzle by drawing on comparative research on policy engagement by charities and using a constructivist interpretation of neo-institutionalism. In the article, we use the term "policy engagement" as a shorthand for the CRA's (2019) term "public policy dialogue and development activities."

The article examines the context for federal policy engagement by all charities in Canada but focuses on charities in the international development sector and specifically the 58 charities that were members of the umbrella organization called Cooperation Canada as of 2018 (see Appendix 1). Charities in different sectors face different constraints and opportunities for policy engagement. The international development sector is an instructive case, if not representative of the full spectrum of charities in Canada, for several reasons. First, most of the large and medium-size organizations in the sector are registered charities, so charity law has an important impact on the operation of the sector. Second, the primary focus of almost all charities in the sector is service provision (see Appendix 1). Such organizations are more likely to have larger budgets than organizations focused primarily on policy engagement, which means that the CRA's 10 per cent cap on "political activities" was less of a constraint than for smaller organizations and those with a strong policy focus. Third, most charities in this sector expressed strong belief in the importance of public policy engagement but struggled to prioritize those beliefs in their operations.⁴ Fourth, charities in the sector asserted that their engagement in public policy advocacy was constrained by the old CRA regulations. A 2016 report on a survey conducted by the Canadian Council for International Cooperation (now Cooperation Canada) stated that "current CRA rules ... appear to limit the willingness of Canadian charities to undertake public policy work" (CCIC, 2016a: 14). Ninety-two per cent of the survey respondents indicated that the old CRA restriction on "political activities" "hinders their ability to realize their missions" (CCIC, 2016b: 23).

In this sense, the international development sector was a "most likely case" to study (see Flyvbjorg, 2011; Ruffa, 2020), as it appeared to be poised to increase its policy engagement following the changes in the ITA and CRA regulations—but that does not appear to have happened. The sector is also instructive because international development is primarily an area of federal jurisdiction, so policy engagement by charities in the sector focuses on the federal government and can be analysed through federal government data, a task that is more complicated for charities engaged with provincial and municipal policy issues for which data are more difficult to compare. Despite the particularities of the international development sector, our interviews with charity lawyers and leaders of organizations that support a wide range of Canadian charities suggest that many of the factors that we observed in the international development charities are also prevalent in other charitable sectors, bringing it close to the definition of a "typical" case (Gerring, 2008; Seawright and Gerring, 2008).

While controversies about the 2006–2015 Conservative government's enforcement of the CRA regulations contributed to widespread perceptions that the regulations themselves were the primary constraint, our research points to a wider range of factors that shape and constrain policy engagement by charities. First, charity leaders highlighted that it was not so much the regulations themselves as the politicized enforcement of the regulations in the last few years of the Harper government that generated fear and uncertainty in the sector about policy engagement. Moreover, that fear and uncertainty and the risk aversion that emerged from it persists within many charities, especially within boards of directors, reflecting socially constructed understandings of the law. Second, other factors that have little to do with the regulatory framework also shape policy engagement by charities in important ways, including the difficulties of finding resources to pay for public policy engagement, concerns about jeopardizing government funding and a strategic preference for insider advocacy—which was never restricted by the ITA or CRA regulations. The insider approach to policy

advocacy focuses on building relationships with decision makers and quiet lobbying behind the scenes, in contrast to outsider approaches involving public campaigns and calls to action, which were regulated and restricted by the ITA and CRA regulations (see DeSantis and Mulé, 2017; Lang, 2012; Sussman, 2007).

The article develops this argument in five parts. Section 1 highlights key debates in the comparative literature about public policy engagement by charities. Section 2 explains our methodology and data sources. Section 3 provides a short history of federal law⁵ on policy engagement by charities in Canada, the responses of charities to the law, and debates over whether and how the ITA and regulations should change. Section 4 presents our analysis of publicly available federal government data on policy engagement by charities. Section 5 explores the constraints on policy engagement in more detail, drawing primarily on interviews with charity leaders.

Literature Review and Theoretical Framework

Whether and how charities engage with public policy is an important issue in any polity. Charities play key roles in modern pluralist societies not only by providing social services but also by monitoring governments, holding them to account, providing policy advice and stimulating public deliberation on policy issues (Gibbons, 2016; DeSantis and Mulé, 2017). Since charities often act as implementing agents for state policy, they have important experiential knowledge about what works and what does not and so can contribute to better policy outcomes.

Public controversies over the old CRA regulations on policy engagement (2003–2018), especially under the Harper government, generated significant analysis of the legal constraints on policy engagement by Canadian charities, in addition to the extensive earlier literature on the 1978 CRA regulations (Blumberg, 2018; Devlin, 2017; Drache, 2002; Elson, 2011; Hale, 2017; McGregor-Lowndes and Wyatt, 2017; Monohan and Roth, 2000; Parachin, 2007, 2015, 2016, 2017a, 2017b; Singer, 2020; Watson, 1985). This analysis, which we explore further in section 3, is extremely helpful for understanding the formal regulatory framework for policy engagement by charities, but little analysis has been published on how charities have responded to the 2018–2019 changes in the law (see Lorinc, 2020). The other contribution of this article is thus to fill that empirical gap in the literature. Since our research findings indicate that the 2018–2019 legal changes did not spur an important increase in charities' policy engagement, research is needed to better understand the broader factors that shape policy engagement by charities, including how charities understand and respond to the law.

A social constructivist approach to institutionalism highlights the ways that actors' understandings of institutions—such as the regulatory framework composed of the ITA and associated CRA regulations—shape their behaviour. Classical versions of neo-institutionalism understood institutions as objective structures that shaped and constrained human action, "analogous to the rules of the game in a competitive team sport" (North, 1990: 4). However, there has been considerable debate over the extent to which institutions should be understood objectively as fixed structures or as contingent social constructs, based on human understandings and perceptions. In this article, we do not engage directly with these theoretical debates, but we do note that scholars on all sides of these debates agree at a basic level that how political agents understand their institutional environments is crucial

to explaining their political behaviour. From this perspective, it is not the law itself that shapes and constrains the behaviour of charities, but rather how charities, their boards and managers understand (and frequently misunderstand) the ITA and the CRA regulations—including the ways that understandings based on previous versions of the law and their implementation under previous governments continue to shape charities' behaviour into the present (see DeSantis and Mulé, 2017: 16–17; and Lasby and Courdeaux, 2016). A social constructivist approach to institutionalism focused on charities' understandings of the law may thus help to solve the apparent puzzle of why public policy engagement by charities does not appear to have changed following the 2018–2019 reforms of the ITA and CRA regulations.

However, while the regulatory framework and charity actors' understanding of it are key parts of the context that shapes policy engagement by charities, comparative research highlights a wider range of factors beyond regulations that may also help to solve the apparent puzzle of policy engagement by charities. Much of the US research emphasizes the ways that resource constraints of money, staff time and expertise limit and shape policy engagement by charities and nonprofits. Serious and consistent policy work costs money—especially to pay for experienced and professional staff—and raising funds to cover those costs is challenging for many charities and nonprofits, frequently limiting their capacity to engage with policy issues (Bass et al., 2007; Lu, 2018; Pekkanen, Smith and Tsujinaka, 2014; Salamon, 2002).

Within the research on resources for policy engagement, numerous studies examine the effects of government funding on decisions by charities about whether and how to engage in policy advocacy. The empirical evidence strongly indicates that receiving government funding is positively correlated with policy engagement, especially when that funding represents less than 50 per cent of a charity's revenue (Bass et al., 2007; Kelleher and Yackee, 2009; Leroux and Goerdel, 2009; Lu, 2018; Mosley, 2011; Moulton and Eckerd, 2012; Neumayr et al., 2013; O'Reagan and Oster, 2012; Salaman, 2002). Resource mobilization theory offers an explanation for this evidence, highlighting the costs involved in policy work and the ways that government funding contributes to the overall capacity of charities for policy engagement (McCarthy and Zald, 1977; Jenkins, 1987). Similarly, the "paradigm of partnership" highlights the ways that relationships with government funders can open pathways for charities to influence policy (Kelleher and Yackee, 2009; Salamon, 2002).

However, comparative research also indicates that government funding frequently softens the advocacy strategies of charities and nonprofits, "channelling" them towards less-confrontational forms of insider advocacy (Chewinski and Corrigall-Brown, 2020; Clément, 2017; Nicholson-Crotty, 2007) or what Onyx et al. referred to as "advocacy with gloves on" (2010: 41). Mosley (2012) found that public funding led charities to "reject confrontational methods and advocate as insiders" (2012: 841) while Lu concluded that "nonprofits with more government funding use more insider strategies to achieve their advocacy goals" (2016: 9). Given the significant federal support for charities and nonprofits in Canada (Clément, 2017), including in the international development sector (Smillie, 2012), it may be that government funding has channeled charities towards insider advocacy in ways that were (and continue to be) more powerful than the constraints on outsider advocacy contained in the CRA regulations.

The studies on the effects of government funding also highlight the strategic or tactical choices of charities about how to engage governments to promote policy

change, particularly whether they adopt insider or outsider approaches to advocacy (DeSantis and Mulé, 2017; Lang, 2012; Sussman, 2017). Government funding may be one factor that informs these strategic considerations, but there may be other factors as well, such as the risk appetite of boards of directors and organizational evaluations of the relative effectiveness of different tactical options.

In the research for this article, we drew on insights from social constructivist approaches to institutionalism and comparative research on policy engagement by charities by asking questions about the factors that shape Canadian charities' decisions about policy engagement, including the ITA and CRA regulations and charities' understandings of them, the challenges of raising funding for policy work, the effects of government funding on decisions about policy engagement and strategic approaches to policy change by the management and boards of charities.

Research Methods and Data Sources

To try to understand whether and how charities' engagement with federal public policy issues changed following the 2018–2019 amendments to the ITA and CRA regulations, we conducted both quantitative and qualitative research. It bears emphasizing that efforts to measure policy engagement by charities with quantitative data require careful interpretation. There are no thorough, systematic or fully objective sources of data on policy engagement by charities in Canada. No single data source captures all the activities that charities understand as part of policy engagement, from policy research to closed-door meetings with officials to public campaigns. Research based on surveys of charities allows for in-depth questions and nuanced data but is typically limited by low response rates and a fixed point in time (see Lu, 2016: table 1). For this reason, we examine publicly available federal government data, which include much larger response rates and can be measured over time but lack the more nuanced questions that survey research allows.

Based on conversations with charity leaders, we identified three sources of federal government data that provide insight into policy engagement by charities: 1) the CRA data on "political activities" for charities; 2) lobbying reports from the Office of the Commissioner of Lobbying of Canada; 3) data on the submission of briefs and witness testimony for two Parliamentary Committees. Below we explain each source in turn.

From 2003 to 2018, CRA regulations required registered charities to self-report on "political activities" in the annual T3010 information return. The data included whether the charity engaged in any "political activities" and how much money it spent on those activities. In response to our information request (submitted on January 11, 2022), CRA provided data from all T3010 returns for the period 2003 through 2018, representing about 75,000 registered charities per year. When the restrictions on "political activities" were abolished in 2019, CRA also stopped collecting data on them, making it impossible to track changes in self-reported "political activities" through CRA data after 2018. The pre-2018 data, however, still provide a useful baseline indicator of charities' engagement in outsider forms of policy advocacy.

Lobbying was never restricted by the CRA but lobbying data from the Office of the Commissioner of Lobbying do provide an indication of insider policy engagement by charities. Federal law requires any individual or private entity who lobbies a federal public office holder regarding possible changes in any federal policy, law or decision to register as a lobbyist and to register each lobbying visit. The data, published on the website of the Office of the Lobbying Commissioner⁹, include the name of the lobbyist and the person or entity being lobbied, the subject matter of the meeting and the date of the meeting.

The House of Commons Standing Committees represent an important opportunity to engage with Members of Parliament on policy issues, despite the contradictions generated by party discipline (Franks, 1971; Skogstad, 1985). Data on who presented to what committee and when are available on the Parliamentary website¹⁰. We examined the lists of all organizations that submitted briefs and witness testimony to the House of Commons Standing Committee on Foreign Affairs and International Development (FAAE) and to the annual Pre-Budget Consultations of the House of Commons Standing Committee on Finance (FINA) over the period 2000–2022. We chose the FAAE and FINA, since these are the parliamentary committees of most relevance to the international development sector.

It is important to recognize that these federal datasets are marked by misreporting, under-reporting, over-reporting and by different understandings of the law (see Blumberg, 2012; Lasby and Courdeaux, 2016). Charities' understandings of the regulations for policy engagement and their risk appetites may have also changed over time; so, apparent temporal patterns may indicate changes in reporting rather than actual increases or decreases in policy engagement. It is also important to highlight that the federal government data do not capture important forms of policy engagement that were not restricted by the 2003-2018 CRA regulations, especially outward facing advocacy such as the publication of policy research, public education, blogs and op-eds that did not involve public calls to action. Because of the limitations of the data, we do not attempt statistical analysis beyond basic description. We use the three sets of federal data as overlapping radar screens to identify which charities have been engaged in federal policy processes at all. We posit that charities that have been seriously and consistently engaged in efforts to influence federal policy will show up in at least some of the data sources, some of the time. Taken together, these three federal datasets will provide important insights into the overall level of policy engagement by charities, and of the identities of those most involved.

To address limitations in the quantitative data and to gain other insights as well, we conducted 90 semi-structured qualitative interviews with relevant staff (for example, those responsible for policy engagement, senior managers) and board members from 56 Canadian charities, primarily in the international development sector. We conducted 79 of those interviews in English and 11 in French, 58 before the legal changes in 2018–2019 and 32 afterwards. We also interviewed four prominent charity lawyers and four leaders of organizations that support policy work across the entire charitable sector following the 2018–2019 legal changes. We then coded interview transcriptions using a priori and inductive approaches to identify key themes (Azungah, 2018).

The "Long Freeze" on Policy Engagement by Charities

Controversy about the regulatory constraints on policy engagement by charities made headlines when the Conservative government of Stephen Harper announced additional funding to the CRA in the 2012 federal budget to conduct special audits on the

so-called "political activities" of Canadian charities. ¹¹ Evidence suggested that the CRA's audits of "political activities" targeted charities with policy positions that challenged the Harper government, particularly in the environmental, international development, antipoverty and human rights sectors (International Civil Liberties Monitoring Group, 2021). The result of the special audits, conducted between 2012 and 2015, combined with direct threats (Clark, 2010) and cuts in funding to charities that had been seen as being opposed to the Conservative government's policies, such as Kairos, North-South Institute and CCIC, was a "chill" on policy engagement by charities—even within the bounds of the CRA regulations (Kirkby, 2014; Beeby, 2014). As numerous charity leaders lamented, "advocacy" became a dirty word and came to be perceived as a risky activity (DeSantis and Mulé, 2017; Byrnes and Palassio, 2022).

However, the advocacy chill of the 2010s marked only a slight drop in temperature within what can be understood as a "long freeze" on policy engagement by charities that profoundly shaped their institutional design, organizational culture, staff capacity and strategic decision-making (Byrnes and Palassio, 2022). The history of the aversion to policy engagement by charities in Canada is partly rooted in the CRA's regulation of charitable and political "purposes," which in turn is based in common law. Based on the 1891 Pemsel case in the UK House of Lords and subsequent case law, which in turn draws on the 1601 Statute of Charitable Uses, the CRA recognizes four "purposes" for charities: 1) the relief of poverty; 2) the advancement of education; 3) the advancement of religion; and 4) other purposes beneficial to the community (CRA, 2013; Pemsel Case Foundation, 2021). Notably, influencing policy is not a recognized charitable purpose in common law and for many decades, the Canadian parliament and courts did not consider policy engagement to be an allowable "activity" for charities, even to achieve recognized charitable "purposes." For example, regulations put in place in 1978 under the then Liberal government allowed charities to engage in quiet lobbying but prohibited virtually all activities to engage the public with government policy, including "writing letters to the editor," "holding a demonstration," "lobbying government through an organized campaign," and "conducting a campaign where people send form letters to their elected representatives" (Department of National Revenue, 1978; Singer, 2020: 688-90). For over two decades (1978-2003), Canadian charities were created, operated and evolved within a regulatory framework that specifically discouraged policy engagement. The boards of directors and management of Canadian charities generally adopted cautious approaches to public policy engagement and very few charities developed strong capacities for policy analysis and influence (see Carter et al., 2005).

The "long freeze" on policy engagement by Canadian charities and calls for legal reform are well-documented in public statements by charity sector leaders, lawyers, judges, members of parliament and multiple rounds of federal commissions and reports, including the National Advisory Council on Voluntary Action (1977), the Panel on Accountability and Governance in the Voluntary Sector (1999), the Voluntary Sector Initiative (2002), and the Consultation Panel on the Political Activities of Charities (2017) (see also Elson, 2011; Drache, 2002; Laforest, 2013; Levasseur, 2012; Monohan and Roth, 2000; Parachin, 2017b; Watson, 1985). Summarizing the long debate over the regulation of policy engagement by charities, Elson (2011: 62) highlights the "eerily familiar" repetition of calls to modernize the definition of charity in Canada and to relax or eliminate the regulatory restrictions on

policy engagement. As the federally-appointed Consultation Panel on the Political Activities of Charities asserted in its 2017 report, "problems with the legislative framework and its administration have left the [charitable] sector and its regulators stuck on a merry-go-round of consultation, clarification, and concern for nearly four decades" (Consultation Panel, 2017: 10).

Following several public controversies and debates in Parliament in the 1980s and 1990s, the CRA introduced a "Policy Statement on Political Activities" in 2003 (CRA, 2003). The guidance recognized the value of policy engagement by charities (section 3, paragraph 5), but simultaneously restricted it. Partisan activities were still prohibited (CRA, 2003: section 6.1). However, the guidance also distinguished between so-called "political" and "charitable" forms of policy engagement. It allowed charities to engage in limited "political activities" provided that they were related to the charity's officially stated purposes, and that "substantially all" of the charity's resources were directed towards "charitable activities" (that is, not "political activities"). The CRA interpreted the term "substantially all" to mean 90 per cent, ¹³ thus allowing charities to allocate up to 10 per cent of their annual revenue to "political activities," a regulation that became known informally as "the ten percent rule." The CRA defined an activity as "political" if it "explicitly communicate[d] a call to political action (that is, encourage[d] the public to contact an elected representative or public official and urge(d) them to retain, oppose, or change the law, policy, or decision of any level of government in Canada or a foreign country)" (CRA, 2003: section 6.2). Behind-the-scenes lobbying for policy change and public campaigns to raise awareness of policy issues were considered charitable, and therefore unrestricted, as long as they did not involve a public call to action to change a law, policy or decision. The definition of "political activity" as a nonpartisan public call to action is significant because it explicitly discouraged efforts to engage citizens in policy debates and implicitly encouraged behind-the-scenes insider forms of policy engagement, an issue we return to below.

In principle, the 2003 CRA guidance on "political activities" opened opportunities for policy engagement by charities. Public campaigns to change government laws and policies were simply limited to 10 per cent of a charity's annual revenue, with higher allowances for smaller charities and the option to average expenditures on "political activities" over three years.

Public concern about the advocacy chill under the 2006–2015 Conservative government was powerful enough that the Liberal Party proposed to end it in its 2015 election platform (Liberal Party of Canada, 2015: 33–34). After the election, the prime minister's mandate letters to the ministers of National Revenue and of Finance explicitly called for them to end the "political harassment" of charities and to clarify "the rules governing 'political activity,' with an understanding that charities make an important contribution to public debate and public policy" (Trudeau, 2015). The Liberal government eventually terminated the CRA's "political activities" audit program in 2017 (Beeby, 2017). In 2016, the government instructed the CRA to establish the "Consultation Panel on the Political Activities of Charities," which conducted hearings across the country and released a report in 2017 calling for the removal of all restrictions on nonpartisan public policy activities (Consultation Panel, 2017). The report of the consultation panel highlighted the "remarkable consisten[cy]" in responses from charities and emphasized the "strong message" that a "lack of clarity, whether with the rules or their

application, means some charities view political activities as too risky to carry out and engage in self-censorship...many charities make a rational choice to avoid or limit the risk" (Consultation Panel, 2017: 12). The government was then forced to act even more quickly when the Ontario Superior Court of Justice (2018) ruled that the CRA regulations on "political activities" were unconstitutional, since they violated the right to freedom of speech. In December 2018, Parliament amended the ITA to remove references to "political activities" by charities, and in January 2019, the CRA issued new policy guidance (CG-027) for unlimited nonpartisan "public policy dialogue and development activities" by charities (CRA, 2019).

Quantitative Data on Policy Engagement by Charities

When we overlap the quantitative data from the three federal data sources, they tell a consistent story: very few Canadian charities engage seriously and consistently with public policy, and there is no evidence that the 2018–2019 changes to the ITA and CRA regulations prompted any significant change in policy engagement by charities in the international development sector.

With over 85,000 registered charities in Canada, the volume of data over 20 years is too large to include here, and so we provide it as supplementary data on the journal website. Here we summarize the highlights.

Canada Revenue Agency data on "political activities"

The CRA data on self-reported "political activities" by registered charities show that very few charities ever reported "political activities" and the amounts of expenditure that they reported were small (see supplementary data, tab 1). Of roughly 75,000 registered charities, an average of just 419 per year (0.57%) ever reported any "political activities." Of the charities that did so, spending on "political activities" represented 0.91 per cent of total revenue, a proportion that declined over time, especially after 2010. Mean spending on "political activities" by the charities that reported them was \$6,514 per annum. ¹⁵

Reported spending on "political activities" by Cooperation Canada member charities was higher than the charitable sector as a whole, but still relatively small and limited to a small group. Of the 58 registered charities that were members of Cooperation Canada in 2018, a maximum of 24 (in 2015 and 2016) ever reported any "political activities," but only a smaller group of 15 charities consistently reported spending more than \$10,000 per year. Spending on "political activities" by the Cooperation Canada charities that reported them averaged 1.36 per cent of total annual revenues over the period 2003–2018, nowhere near the 10 per cent cap. Moreover, many of the managers of international development charities that reported spending on "political activities" indicated in interviews that they did so out of an abundance of caution and included spending on policy activities outside the CRA's definition of "political activities." So, the number of international development charities that used public calls to action in their efforts to influence federal policy between 2003 to 2018 was likely even smaller than the CRA data indicate.

Lobbying reports submitted to the office of the Commissioner of Lobbying of Canada

The data on lobbying activity by international development charities paint a similar picture to the CRA data. Between 2008 and 2022, 25 of the 58 Cooperation Canada

charities submitted one or more lobby reports, with a maximum number of 24 organizations reporting in 2022. Of those, a smaller group of 15 charities submitted lobby reports on a regular basis (an average of more than one per month).

The number of reports submitted by Cooperation Canada member charities increased slightly (4.7%) in the three years following the change in the ITA and the CRA regulations (2019–2021) compared with the three years before the change (2016–2018), while the number of Cooperation Canada charities that submitted lobby reports increased from 17 in 2016 to 22 in 2021. However, as we discuss further below, the leaders and staff of international development charities did not indicate in our interviews any increases in policy engagement over that time period. So the data from the lobbying commissioner may indicate increased rates of reporting rather than increased policy engagement; they may also indicate increased rates of lobbying for funding in the early years of the new Liberal government, which was widely seen as more supportive of the sector than the previous Conservative government.

House of Commons standing committees

Between 2000 and 2022, over 15 sessions of Parliament, the House of Commons Standing Committee on Foreign Affairs and International Development invited a total of 2,962 witnesses and received 464 written briefs. Of those, 196 witnesses (6.6%) and 31 briefs (6.6%) were from 28 charities belonging to Cooperation Canada. However, just 8 Cooperation Canada member charities engaged with the FAAE ten times or more, while another 8 engaged only once or twice over the 23-year period.

Between 2000 and 2022 the House of Commons Standing Committee on Finance produced 16 pre-budget consultation reports, involving a total of 4,680 witnesses and 6,833 briefs. Of those, 25 witnesses (0.53%) and 52 briefs (0.8%) were from 17 Cooperation Canada charities. Only 3 Cooperation Canada charities engaged with FINA ten times or more, while 9 engaged only once or twice over the 22-year period. With few exceptions, these are the same charities that appear in the data from CRA and the Lobbying Commission. The pattern of engagement with the 2 standing committees did not change significantly after 2018: 7 charities did submit briefs to FINA for the first time after 2019, although 5 of the 7 did so only once and none were invited to testify. Only 1 charity engaged with the FAAE after 2019 that had not done so before, while 11 charities that had submitted briefs or testified to the FAAE before 2019 did not do so between 2019 and 2022.

Explaining Policy Engagement by Canadian Charities

The quantitative data from federal sources on policy engagement by charities point to three basic observations: 1) relatively few charities have consistently engaged with federal policy issues; 2) the charities that have engaged with policy spend proportionately few resources on it; and 3) there is very little evidence of any change in policy engagement following the 2018–2019 changes to the ITA and CRA regulations—either in the number of charities or the frequency of their engagement. In this section, we explore factors that might explain the apparent lack of change in policy engagement by charities after 2018–2019, based primarily on our qualitative interviews.

Charity leaders in the international development sector whom we interviewed asserted almost universally that the change in regulations "did not make any difference" for decisions by their organizations about whether, how or how much to engage with

federal policy issues (Respondent 81, 2023). They explained that, as organizations with relatively large budgets and a focus on service provision, expenditures on "political activities" were "never anywhere near ten percent" (Respondent 65, 2022). One policy director indicated that "even with a big campaign every year and two full time staff salaries, we still don't get anywhere near the ten percent threshold" (Respondent 15, 2016).

Nevertheless, charity leaders did emphasize that they appreciated the change in the ITA and the CRA regulations, partly because the new regulations removed the "annoying and time consuming" reporting requirements (Respondent 67, 2022) and partly because they removed "fear and uncertainty" and "feelings of being watched by the CRA" (Respondent 71, 2022). Some respondents complained that the 2003–2018 CRA guidelines were "vague and confusing" (Respondent 43, 2016), which discouraged many smaller charities from engaging with policy issues because they could not afford legal advice or dedicate staff time to overcoming the confusion. At the same time, charities with a strong commitment to policy engagement reported that, despite the confusing guidelines, "we figured out how to work within the rules" (Respondent 37, 2016).

The party in power and the politicization of CRA regulations

Rather than the CRA regulations themselves, charity leaders considered the politicization of the regulations by the Harper Conservatives to be a much more important constraint on policy engagement, especially between 2012 and 2015. As one policy director put it, "the CRA audits under Harper were targeted, deliberate and politically motivated" (Respondent 38, 2016). Another asserted, "The Income Tax law itself is not the problem, but its application... it's the harassment and political targeting of charities that was the problem" (Respondent 7, 2016). Charities that were audited by the CRA emphasized "traumatizing transaction costs" (Respondent 29, 2016). As one former executive director explained, "the (CRA) audit went on forever, they wanted to access everything, transcripts of public talks, speaking notes, everything, it was nuts! It was incredibly time consuming, huge staff energy and most of it fell on me" (Respondent 84, 2024). Stories of CRA audits spread quickly through the charity sector and generated widespread fear of the audits themselves and of the even greater risk of losing charitable status and the associated fundraising capacity.

Major cuts in government funding to prominent international development charities that had been seen as being opposed to the Conservative government's policies, such as CCIC, Kairos and the North-South Institute, further heightened the sense of fear and subsequent "chill" on policy engagement (Beeby, 2014). One seasoned policy advocate explained the widespread perception that "if you spoke up, you were cut off. Eventually, people started silencing themselves" (Respondent 46, 2017). Another reported that "discussions [about policy engagement] within my organization and the international development sector were characterized by panic and fear" (Respondent 19, 2016). Other charity leaders highlighted the practical effects of the "chill" on policy engagement: "it made you hesitant to speak out in public" (Respondent 25, 2016); "You could feel the chill, it was palpable, nobody even wanted to write public letters or op-eds" (Respondent 84, 2024). A government relations director from a large charity reported that "our board didn't even want us to do a pre-budget submission [to the Standing Committee on Finance] or other totally

legitimate policy activities that were not even restricted by the CRA" (Respondent 27, 2016). The leader of another policy team explained that "the perception of risk [about policy engagement] by the board emerged at a percentage point well below ten percent" (Respondent 31, 2016), suggesting again that the 10 per cent rule itself was not a constraint but rather perceptions of its politicized application by the Harper government. In the context of those perceptions, any uncertainty or confusion about the CRA rules contributed to risk aversion: "sometimes we'd forget what was allowed and perfectly legitimate. The chill created doubt and restraint on our part about what messages to send" (Respondent 19, 2016).

Unprompted by our interview questions, many charity leaders said that the change in tone of the new Liberal government starting in 2015 was more important than the change in CRA regulations. One policy officer explained, "the [Liberal] government perceives the role of civil society very differently. They actually call us to ask what we think" (Respondent 1, 2016). Others emphasized that "our access to the government opened up" (Respondent 64, 2022), and "the stress of cuts, the advocacy chill was lifted" (Respondent 45, 2017). Charity leaders suggested that the attitude of the government in power may be a more important factor in determining policy engagement than the law itself.

The greater openness to policy engagement by charities under the Liberal government prompted some organizations to invest new resources and hire new staff for policy research and advocacy, but few reported any significant changes in the strategies they used to try to influence policy making. Most notably, charity leaders indicated a strong ongoing preference for insider strategies, and none reported that they had increased "public calls to action" despite the 2018–2019 changes to the ITA and the CRA regulations. As one executive director explained, "the [Liberal] government has generally been good for this sector . . . they've supported us, so I don't want to spit in the soup . . . When you have a friend, it means that sometimes you have to talk the real talk and say 'Look, I'm not happy with this', but it's a soft-spoken advocacy toward the current government" (Respondent 82, 2023).

However, charity leaders also revealed some deep and persistent legacies of the "chill" on public policy engagement, even eight years after the change in government and five years after the change in the law. As one executive director explained, "Many charities lost their capacity for advocacy. Advocacy units were shut down or downsized and staff were redeployed. It takes longer for organizations to re-build, to hire new staff, to develop new strategic plans. Those ten years [of Conservative government] were corrosive and hollowed out the capacity for advocacy in many charities" (Respondent 52, 2018). While some organizations have managed to hire new staff for policy research and advocacy, feelings of uncertainty, fear and risk aversion about policy engagement persist throughout the sector—highlighting the insights from constructivist institutionalism that what matters is not simply the regulations but also how they are understood. Staff from charities with active histories of policy engagement reported that within their own organizations "there is still a level of alarm when the word 'advocacy' is used" (Respondent 75, 2023) and "the fear is so ingrained that it [advocacy] still feels a little risky" (Respondent 70, 2022). Another explained that risk aversion to policy engagement "still continues like a cloud over civil society. It [policy engagement] is like a muscle we haven't used yet...there's still fears around it" (Respondent 68, 2022).

Mistrust and fear of the CRA

Charity leaders indicated that their fear is partly grounded in an ongoing mistrust of the CRA that is pervasive across the charitable sector (see Advisory Committee on the Charitable Sector, 2021). Some policy directors indicated that they did not feel that the 2019 CRA regulations had yet been seriously tested and they were still waiting to see how the CRA and the rest of the federal government will respond to charities that engage aggressively in public calls to action for policy change. One executive director explained, "being first over the trench is scary" (Respondent 76, 2023). The charity lawyers we interviewed explained that, despite the change in the law, charities still have good reasons to fear the CRA-in particular, the onerous transaction costs and reputational damage incurred through CRA audits (see Carter, 2015) and the high cost of appealing CRA decisions (see Chan, 2016; Juneau, 2016: 2; Senate of Canada, 2019: 69–73). The absence of an affordable process for appealing CRA decisions led one charity lawyer to describe the CRA as "beyond the rule of law" (Respondent 74, 2023). In the context of legal advice that still often views policy engagement as a risk, many charities continue to avoid any activities they think might attract the CRA's attention.

Charity leaders were also concerned that future federal governments might re-politicize the CRA audit process, either by reintroducing legal restrictions on policy engagement or by "weaponizing" the CRA's complaint mechanisms and audit power to punish outspoken charities and to frighten others into silence (Respondent 80, 2023). In this context, some respondents reported that their organizations remain wary of investing energy and resources to expand their policy engagement, even if they had the money to do so.

Nonregulatory constraints on policy engagement

Our interviews with charity leaders also suggest that decisions about whether, how and how much charities engage in policy debates are heavily shaped by a combination of organizational culture, leadership, strategic considerations and funding—reinforcing findings from comparative research, albeit with some important nuances. For most of the charities we interviewed, these nonregulatory factors did not change after 2018–2019, thus offering insights into the lack of changes in policy engagement.

Organizational culture as a constraint on policy engagement

Senior managers highlighted multiple factors that draw their attention and support away from policy engagement. Those factors included the tension between the long-term and uncertain results of policy engagement and the organizational focus on short-term results fostered by the results-based management frameworks required by governments and philanthropic donors (Respondent 35, 2016). The director of one charity explained, "The metrics that [senior managers] are judged on are deliverables, impacts and outcomes... which doesn't lend itself to long-term thinking about advocacy for system change" (Respondent 53, 2018). Another executive director highlighted the ways that the constant demands of program management and fundraising distracted his attention from policy engagement: "you're distracted, you're focused on those other things, program implementation and

funding" (Respondent 73, 2022). Reinforcing social constructivist approaches to institutionalism, many policy directors also emphasized misunderstandings by board members about CRA regulations as a constraint on policy engagement. As one explained, "Boards have the sense that the rules don't allow 'political activities' at all so they won't even try. They don't take risks" (Respondent 53, 2018).

Several executive directors and policy directors emphasized that for charities to engage with policy, their leaders must actively prioritize and nurture it. As one director put it, "if you want to do effective public policy advocacy, you must staff it and you must resource it . . . that helps to create an internal culture, an identity as an organization that does public policy advocacy. It's not just a question of the resources but also the identity that this is something that you do and it's resourced, it's part of your priorities as an organization" (Respondent 64, 2022). Another director spoke with praise about the principled commitment of another charity that "made really clear, strategic decisions to build a policy team instead of spending unrestricted money on fundraising campaigns" (Respondent 71, 2022). Many policy directors also emphasized their ongoing work to "advocate for advocacy" within their own organizations and the need to regularly remind their directors and boards of the value and positive results of policy work (Respondent 75, 2023). One policy director explained that "I had to do a lot of internal advocacy to get policy advocacy into the strategic plan" (Respondent 78, 2023). Another explained that "a big part of my job is to convince [the executive director] and the board that policy work generates results and does not pose a risk to our funding" (Respondent 65, 2022). Both before and after the 2018-2019 changes to the ITA and the CRA regulations, interviewees gave similar accounts of how organizational culture affected their organization's approach to policy engagement.

Strategic constraints and an ongoing preference for insider advocacy

Within the charities with at least some policy engagement, policy staff were adamant that decisions about how to influence government policy-and especially whether to use insider or outsider approaches—were driven primarily by strategic considerations about what types of engagement are likely to be most effective. Again, this finding came out in interviews before and after the 2018-2019 legal changes. As one seasoned policy director put it, "the only constraints are strategic" (Respondent 10, 2016). Many respondents emphasized their efforts to "use our budget and resources as strategically as possible" (Respondent 79, 2023) and "on topics that have the most chance of getting somewhere" (Respondent 13, 2016). Policy leaders also concurred on the "need to pick a few issues and really push those few" (Respondent 68, 2022) and to focus on winnable fights. As the policy director for a charity with a relatively large budget for policy engagement explained, "We don't advocate on hopeless issues. We would be wasting valuable political capital to push for issues the government is not interested in. We focus only on issues on which we have a chance of success" (Respondent 31, 2016). A government relations consultant pointed to the importance of tackling "low hanging fruit to help ensure the success needed to maintain board enthusiasm [for policy engagement]" (Respondent 44, 2016). Similarly, a policy director explained the ways that prevailing results-based management frameworks push charities towards short-term, winnable policy issues: "An advocacy campaign built around an idealistic stand won't enable you to report on the results the way that [our management framework] requires" (Respondent 34, 2016). However, the same policy director also recognized the tradeoffs of "a focus only on what you can win rather than what is important."

Both before and after the 2018–2019 changes in the law, almost all the charity leaders we interviewed expressed a strong strategic preference for insider approaches to policy engagement. It is important to highlight that the CRA restrictions on "political activities" from 2003–2018 applied only to outsider forms of policy engagement involving a public "call to political action" (CRA, 2003: section 6.2). Insider forms of advocacy such as lobbying and public engagement that did not involve a public call to political action were not restricted. The widespread strategic preference of charity leaders for insider advocacy thus helps to explain why the 2018–2019 changes in the ITA and CRA regulations has had little apparent effect on policy engagement by charities.

Policy directors explained that they preferred insider approaches both because they typically require less time and resources than public campaigns and because they contribute to building relationships of trust with decision makers, which they considered to be essential for long-term policy influence. Some charities indicated that if insider approaches failed, they would try other, more aggressive, public-facing tactics. As a policy director for one of the more activist charities explained, "We always try the insider track first. If we're getting there with the insider track, and the government comes through with our priorities, we're happy to leave it at that. But if it doesn't happen, then we try something different" (Respondent 70, 2022). Another said, "You always try to get what you want through the back channels first, without embarrassing someone" (Respondent 53, 2016).

However, the policy directors for many charities explained that they exclusively use insider approaches focused on relationship building and never engage in campaigns involving public calls to action. As the policy lead for a large coalition organization explained,

Our advocacy approach is very much a relationship-based approach. It's behind the scenes, direct advocacy with those who we assess to be key decision makers on whatever issue we're trying to push forward...we're not a campaign style organization. It's not that we are never critical, but we never threaten, we never seek to embarrass. We're more apt to try to find mutual win-win solutions. (Respondent 83, 2023)

The policy director for a large charity similarly emphasized, "we don't come out to criticize, it's not our style. We do things behind the scenes. We build long-term relationships with people in government" (Respondent 13, 2016). Policy directors also explained the importance of "collaborative, solutions-focused" approaches to policy engagement as essential for building trusting relationships with government decision makers (Respondent 78, 2023). This approach involves "being a helpful partner to government" (Respondent 38, 2016) and "propositional, not oppositional" (Respondent 31, 2016). A seasoned government relations consultant emphasized the importance of making proposals that government decision makers "can say yes to" (Respondent 76, 2023). Another policy director similarly explained,

"good advocacy requires engagement with governments that want praise and recognition for their work" (Respondent 10, 2016).

While most of our respondents claimed to respect outsider forms of policy engagement by other organizations, they did not consider outsider approaches to be appropriate strategies for their own organizations. One explained, "I don't think that putting the government on blast and telling them all of the things they're doing wrong is going to get you where you want to go" (Respondent 83, 2023). Another asserted, "I don't believe that raising your fist in the air really effects change. Rather, advocacy is about being really smart about using media at the same time as pressuring on the inside. At some point, everybody has to win, including the decision makers you are lobbying" (Respondent 46, 2017).

Only two of the charities we interviewed rejected the insider approach to policy engagement. Interestingly, even these charities acknowledged that they considered insider advocacy to be more effective for achieving specific policy goals. As one put it, "closed door advocacy is often more productive, takes things further" (Respondent 50, 2018). Their rejection of insider approaches was principled rather than strategic, based on "strong lines of accountability to partner organizations in the global South" (Respondent 29, 2016) and "a conscious choice to engage citizens in the policy process" (Respondent 50, 2018). As Respondent 29 questioned "if you're only doing the inside strategy, what happens to public engagement, to public education, to movement building?" (Respondent 29, 2016). The director of the other charity explained her organization's principled commitment to citizen engagement in policy change: "Policy interventions without public engagement may shift policy, but they don't shift power. We're aiming for more than just technical policy changes" (Respondent 35, 2016). She went on to explain her organization's rejection of insider advocacy:

Instead of analyzing the situation and saying, "this is what we want", there's a shift to asking, "in this political context, what can we realistically achieve?" It's a shift from the clear and principled to the tactical and strategic. People constrain their thinking to the political context and don't ever think big picture, they focus only on what they think they can get in this political moment. We thus confuse the winnable achievement with what we really want. (Respondent 35, 2016)

Difficulties of engaging Canadians with public policy

While some policy directors at charities that prioritized insider approaches sympathized with concerns about the lack of citizen engagement and movement building, they also highlighted the difficulties of engaging Canadians in campaigns to change public policy, particularly on global issues. Again, the findings from interviews were virtually the same before and after the 2018–2019 changes to the law. Policy staff from a wide range of organizations echoed the assertion that "Canadians are generally reluctant to act politically" (Respondent 11, 2016) and "would rather write cheques than a letter to their MP" (Respondent 34, 2016). Seasoned policy advocates asserted that outsider campaigns on international development issues typically generate little public interest: "You get very little uptake. It's a small cast of characters" (Respondent 40, 2016). Policy leaders from two organizations with

large member-donor bases indicated that only 1 to 5 per cent of their regular donors could be engaged in public policy campaigns (Respondents 7 and 13, 2016). As one put it, "it's a real, real challenge to move people beyond charity" (Respondent 13, 2016). The policy lead for another charity lamented that "our organization just doesn't have the resources to move the needle on public opinion" (Respondent 83, 2023). In the context of the perceived challenges of engaging Canadians in efforts to change public policy, all but a few charities with deep, principled commitments to citizen engagement asserted that insider approaches to policy change were a better use of their limited resources.

Receipt of federal funding as a constraint on policy engagement

At the same time, many charity leaders also acknowledged that the receipt of federal funding made them more careful about policy engagement. Echoing the findings from quantitative research in the United States and channelling theory, our interviewees were adamant that reliance on government funding did not stop them from policy engagement but did lead them towards insider approaches and greater caution with outsider approaches, under both Conservative and Liberal governments. Once more, the interview data on this question did not vary much from the pre to post 2018-2019 period. As one policy director explained, "we do receive quite a lot of money from the government and so we always have to calculate how much risk we want to take" (Respondent 70, 2022). At the same time, there is a widespread perception within the sector that the more radical organizations do not receive government funding and so are not constrained by it. As the policy director from an organization with relatively little federal funding explained, "we don't have big programs with [Global Affairs Canada] . . . so we really don't care if we upset the government because there is little risk. It allows for a more outsider, confrontational approach" (Respondent 50, 2018).

Three points about the relationships between government funding and policy engagement bear emphasizing, however. First, almost all of the charities we interviewed received at least some federal funding (see appendix 1). Second, our interviews and the quantitative data indicate that almost all of the charities that received federal funding also engaged with federal policy issues in at least some fashion. And third, there is no obvious evidence that the absence of government funding leads organizations to be more engaged or more outspoken on policy issues. Indeed, some of the more engaged and more critical international development charities have long histories of significant government funding (such as Inter Pares and the two Oxfams), while some organizations with very little or no government funding (such as Kentro Christian Network) have been largely disengaged from public policy debates. Our interviewees both before and after the 2018-2019 changes attributed these variations to organizational missions, culture and leadership within charities and insisted there is no direct relationship between government funding and avoidance of policy engagement. In a separate project, analysis of CRA data for all Canadian charities between 2003 and 2018 pointed to a similar conclusion: Charities without government funding were no more likely to engage in political activities than charities with government funding, even when it represented more than 50 per cent of their revenue (Cameron et al., 2024).

Funding constraints on policy engagement

Charity leaders highlighted the availability of funding for policy engagement as a major constraint that existed before the 2018–2019 legal changes and afterwards. As advocates of resource mobilization theory emphasize, public policy advocacy has "significant costs" (Salamon, 2002: 20), especially the salaries for experienced staff members. Many of our respondents emphasized the challenges of raising funds for public policy work: "It's really difficult to fundraise for advocacy" (Respondent 70, 2022). Given that the federal government does not allow organizations to use federal funds for advocacy towards the government, the two main alternative funding sources are philanthropic foundations and individual charitable donations, neither of which are easy to access and both of which come with strings attached (Chewinski and Corrigall-Brown, 2020).

Funding from philanthropic foundations for public policy engagement in Canada is very limited, despite signs of growing interest (Elson and Hall, 2016; Lauzière, 2021; Pearson, 2022). Many of our respondents pointed to this constraint: "How on earth are we going to get Canadian philanthropies to support us? We have not cracked it... We have not been able to facilitate a conversation with Canadian philanthropists about funding policy engagement in the international development sector" (Respondent 64, 2022). The one significant exception is the US-based Bill & Melinda Gates Foundation, which is the primary source of foundation funding for policy advocacy in the international development sector in Canada. While nine Cooperation Canada member organizations were able to access Gates funding for policy work (totalling at least \$14.6 million from 2014–2022), those funds are also directed towards specific policy issues and require reporting on short to medium term outcomes, which can limit the scope of policy work.

Individual donations to Canadian charities are at historic lows (Canada Helps, 2023) and fundraising for public policy work is particularly difficult. As one policy director explained, "It's a really hard sell to make public policy work sound interesting to donors. It feels less immediate . . . The measurables are also more difficult. It's very hard to show that funds for advocacy actually have impact. Policy shifts can take decades. People don't want to give money for hypothetical policy wins decades down the road" (Respondent 49, 2018). Another argued somewhat more optimistically that while some donors will support policy work, the international development sector as a whole has "not yet realized the potential for advocacy-related fundraising" (Respondent 55, 2018). Charitable donations are also no panacea for policy engagement. Some policy directors reported that they felt just as constrained by their individual donors as by federal funding. One highlighted the challenge of "being true to what our partners in the global South are asking us to do, without losing our supporters here [in Canada]" (Respondent 34, 2016), while another noted "the fine line between what our partners [in the global South] want us to do and what our donor and member base [in Canada] will support" (Respondent 55, 2016).

In sum, the overall lack of resources to support policy engagement continues to operate as a major constraint on charities' investments in it, despite the 2018–2019 legal changes. In this context, the primary funding pattern for policy engagement by charities is relatively large-service delivery organizations that allocate a very small proportion of their total revenue to public policy work—typically drawing

on "unrestricted funding" from individual charitable donations. Some charities have also been able to supplement those unrestricted funds with philanthropic grants, most notably from the Bill & Melinda Gates Foundation—but for most these still represent a small proportion of their overall budgets.

Coalitions as a strategy for collaboration and avoiding policy engagement

In the context of scarce resources and ongoing risk aversion, many charities emphasized in interviews both before and after the 2018-2019 legal changes that they do not engage with federal public policy issues on their own but rather through coalitions of like-minded organizations. Advocacy coalitions enable their members to pool resources and also provide political cover on controversial issues (see Corrigall-Brown, 2022: 54-84; Levi and Murphy, 2006; Staggenborg, 1986). 17 Indeed, many of our respondents highlighted sectoral collaboration through coalition organizations as a "critical success factor" for public policy engagement (Respondent 81, 2023) but also lamented that the sector does not collaborate more in efforts to influence federal policy. The typical coalition arrangement involves individual member organizations that provide funding, and a coalition office that provides research, campaign planning, coordination and public-facing policy advocacy. In that arrangement, the role of individual organizations varies widely, from only making financial contributions to participation in insider lobbying to outsider campaigning. Nevertheless, many of our respondents implied what one policy director acknowledged openly: "we let them [the coalition office] do the heavy lifting" (Respondent 75, 2023). In this context, some charities in the international development sector support policy engagement through financial contributions to coalitions but do not actively engage in efforts to influence policy on their own, especially through public calls to action and other outsider tactics.

The uneven distribution of work between coalition offices and member organizations frustrated some coalition staff and charity leaders. As the sole staff member of one coalition organization explained, "It can't be just the coalitions that do all the work. Coalitions can do the background research and advocacy coordination, but the members need to engage the broader public and the politicians. The member organizations are much better known (to the public), but sometimes they won't speak out" (Respondent 52, 2018). The director of one of the more outspoken charities similarly complained that in some coalitions it is "only a few that are willing to speak to the media" (Respondent 82, 2023).

Coalition funding can also be a constraint. Apart from Cooperation Canada and CanWaCH, which both have mandates that extend well beyond policy engagement, most coalition organizations are very small—often with only enough funding for one staff member. As another charity director acknowledged, "None of the advocacy coalitions have real resources or institutional stability" (Respondent 10, 2016). When coalitions are well-funded and member organizations participate actively in coalition-led campaigns, evidence suggests that they can be very effective strategies for policy change (Corrigall-Brown, 2022). Unfortunately, the opposite pattern of modest funding and passive engagement by member charities in policy coalitions appears to be widespread, with coalitions serving as much to mitigate the perceived risks of policy engagement as to increase effectiveness.

Conclusion

Five years after the ITA and CRA regulations changed to allow Canadian charities to engage in "unlimited" nonpartisan public policy activities, it appears that few organizations in the international development sector have taken advantage of this opportunity. Our research indicates that only a small group of international development charities ever engaged seriously in efforts to influence federal public policy, and that neither the number of charities nor the intensity of their policy engagement changed significantly following the 2018–2019 changes in the ITA and the CRA regulations. Interviews with charity leaders point to at least three interconnected reasons for the apparent lack of change in behaviour by charities despite the change in the law.

First, the CRA regulations on "political activities" from 2003 to 2019 had restricted outsider forms of public policy engagement that involved calls to public action for the purpose of changing laws, policies or decisions of governments. However, almost all the charity leaders we interviewed indicated that their organizations avoided outsider approaches to policy engagement, not only because of the CRA regulations but also because they believe insider approaches are more effective and less likely to jeopardize federal government funding, which almost all of the larger charities in the international development sector receive.

Second, charity leaders also emphasized that it was not so much the ITA and the CRA regulations themselves that had generated confusion and fear among charities, but rather the perception that the Conservative government of Stephen Harper had politicized the CRA regulations by targeting charities that challenged government policies with special audits coupled with cuts in funding to organizations that were seen as opposed to the Conservative government's policies. The 2018–2019 changes in the law did not alleviate concerns that a future government might reintroduce restrictions on policy engagement or simply intimidate outspoken charities by weaponizing the CRA's audit powers and/or cutting funding to charities that challenge government policies. In the context of those concerns, many charity leaders indicated that they were reluctant to make new investments in public policy capacity or engage in outsider strategies. Similarly, charity leaders highlighted that misunderstanding, confusion and uncertainty persist about the ITA and the CRA regulations and broader attitude of the federal government towards policy engagement by charities, especially among boards of directors. In the context of that uncertainty, charities generally seek to mitigate the perceived risks of being audited, losing charitable status or losing government funding by avoiding policy engagement altogether, opting for less risky insider approaches and/or engaging in public advocacy largely under the cover of coalitions. These responses by charities reflect constructivist approaches to institutionalism by calling attention not simply to the laws but also to the ways that they are understood by political actors, including those within civil society.

Finally, the charity leaders we interviewed both before and after the 2018–2019 changes in the law highlighted important nonregulatory factors that shape their decisions about whether, how much and how to engage with government policy. These included the strategic preference for insider forms of policy engagement and related concerns to not jeopardize government funding, as well as the serious

challenges of raising money to support public policy engagement, and the perceived difficulties of engaging Canadians in public policy campaigns.

Controversies over the 2003–2018 CRA regulations on "political activities" by charities created a widespread impression that the laws themselves were the primary constraint on public policy engagement by Canadian charities. However, our research comes to a different conclusion. Our interviews with charity leaders suggest that socially constructed understandings of the regulations and nonregulatory constraints related to strategy, funding and organizational culture are more important in shaping public policy engagement by charities than the formal regulatory framework created by the ITA and the CRA regulations. Understandings of the law have been slow to change and many of the nonregulatory constraints have not changed at all, thus providing an answer to the puzzle of why charities' behaviour does not appear to have changed following a purportedly important change in the law.

Supplementary Material. The supplementary material for this article can be found at https://doi.org/10.1017/S0008423924000623.

Notes

- 1 This research was funded by a SSHRC Insight Grant (435-2020-1036). The authors wish to thank the CJPS anonymous reviewers and participants at the 2023 meetings of the Canadian Political Science Association and Association for Nonprofit and Social Economy Research for helpful feedback.
- 2 The "new" CRA guidance CG-027 was published on January 21, 2019 (replaced with a lightly revised version on November 27, 2020). It replaces the "old" (2003) CRA guidance on "Political Activities" (CPS-022).
- 3 We use the term "political activities" in quotation marks throughout the text to reference the CRA's definition of the term in its 2003 policy guidance, which distorted common understandings of the English word "political" by including all activities intended to engage the public in public policy–making processes but excluding policy activities that did not involve calls to public action.
- 4 A 2016 survey conducted by the Canadian Council for International Cooperation (CCIC; now Cooperation Canada) found that 96 per cent of its organizational members believed it was "very or somewhat important that charities issue calls for action to engage Canadians in public policy debates" but only 30 per cent indicated that they actually did so (CCIC, 2016a: 18, 26).
- 5 In Canada, law includes both statute law (for example, the ITA) and regulations.
- 6 Our research focuses on registered charitable organizations, which are authorized by the CRA to issue federal income tax deductible receipts for donations—which is essential for the fundraising capacity and institutional survival of many organizations. To qualify for charitable status, civil society organizations must comply with the ITA and extensive CRA regulations and reporting requirements. Nonprofit organizations without charitable status are not required to comply with CRA regulations nor can they issue tax deductible receipts for donations (see CRA, n.d.; Imagine Canada, n.d.).
- 7 See the exchanges between Bell (2012) and Schmidt (2017) and between Hay (2017) and Larsson (2018) for a flavour of the debates about constructivist and discursive approaches to neo-institutionalism.
- 8 We excluded public and private foundations from our analysis as their missions make them much less likely to engage in public policy advocacy.
- $9 \ See \ https://lobbycanada.gc.ca/app/secure/ocl/lrs/do/clntSmmrySrch.$
- 10 See https://www.ourcommons.ca/PublicationSearch/en/?PubType=40017.
- 11 The 2012 federal budget provided \$8 million over two years to the CRA to enhance its capacity to conduct audits of "political activities." This amount later increased to \$13.4 million over four years (Beeby, 2014).
- 12 See Parachin (2007, 2015) on the history of "political purposes doctrine" in common law.
- 13 On legal interpretations of the term "substantially all" (which range from 50% to 90%), see Grant (2015).
- 14 The CRA guidelines allowed small charities to allocate up to 20 per cent of revenue to political activities and to exceed that amount if the average over three years remained lower than 20 per cent per year (CRA 2003: section 9 and 9.1).
- 15 Dollar amounts are based on data provided by the CRA and are not adjusted for inflation. Inflation averaged around 2 per cent per annum between 2000 and 2020, with a minimum of 0.30 per cent and a

maximum of 2.91 per cent per annum; see https://www.macrotrends.net/global-metrics/countries/CAN/canada/inflation-rate-cpi.

16 Gates Foundation data indicate a total of 345 grants to Canadian organizations between 1999 and 2024, totalling CAD\$871 million, primarily for activities related to global health and agricultural research and programming. Of that amount, 39 grants (totalling CAD\$62 million) went to 9 Cooperation Canada members—of which 20 grants (totalling CAD\$14.6 million) specifically mention "advocacy" or "policy" activities, all between 2014 and 2022. See: https://www.gatesfoundation.org/about/committed-grants?country=Canada.

17 The most prominent coalition organizations in the international development sector include Cooperation Canada (102 member organizations), Canadian Network for Women and Children's Health (CanWaCH) (103 member organizations), Canadian Network for Corporate Accountability (41 member organizations), Mining Watch (25 member organizations), and the Bigger than Our Borders Campaign against cuts to Official Development Assistance (102 member organizations).

References

- Advisory Committee on the Charitable Sector. 2021. Report #3 of the Advisory Committee on the Charitable Sector. Ottawa: Government of Canada.
- Azungah, T. 2018. "Qualitative research: deductive and inductive approaches to data analysis" *Qualitative research journal* **18** (4): 383–400.
- Bass, G. et al. 2007. Seen but Not Heard: Strengthening Nonprofit Advocacy. Washington, DC: The Aspen Institute. Beeby, D. 2014. "Canadian Charities Feel 'Chill' as Tax Audits Widen into Political Activities." Toronto Star. https://www.thestar.com/news/canada/2014/07/10/canadian_charities_feel_chill_as_tax_audits_widen_into_political_activities.html (July 10).
- Beeby, D. 2017. "Political Activity Audits of Charities Suspended by Liberals." *CBC News.* https://www.cbc.ca/news/politics/canada-revenue-agency-political-activity-diane-lebouthillier-audits-panel-report-suspension-1.4099184 (May 4).
- Bell, S. 2012. "Where Are the Institutions? The Limits of Vivien Schmidt's Constructivism." *British Journal of Political Science* 42 (3): 714–19.
- Blumberg, M. 2012. "How Accurate Are the T3010 Charity Returns When It Comes to 'Political Activities'?" https://www.canadiancharitylaw.ca/blog/how_accurate_are_the_t3010_registered_charity_information_returns/
- Blumberg, M. 2018. "Canada Without Poverty vs. Attorney General of Canada A Pyrrhic Victory for CWP and a Disaster for the Charity Sector." https://www.canadiancharitylaw.ca/uploads/Canada_Without_Poverty_vs._AG_of_Canada_-a_pyrrhic_victory_for_CWP_and_a_disaster_for_the_charity_sector.pdf
- Byrnes, E. and C. Palassio. 2022. "50 Years, 50 Moments (Part 1)." The Philanthropist. March 30.
- Cameron, J., H. Dicks and L. Swiss. 2024. "Biting the hand that feeds you? Exploring whether (and how) government funding constrains charities' engagement in public policy." Paper presented to the Association of Nonprofit and Social Economy Research (ANSER), Montreal, June 12–14.
- Canada Helps. 2023. 2023 Giving Report. https://www.canadahelps.org/en/the-giving-report/
- Carter, S., B. Plewes and H. Echenberg. 2005. "Civil Society and Public Policy: A Directory of Non-Profit Organizations Engaged in Public Policy." Edmonton: Muttart Foundation, J.W. McConnell Family Foundation, Maytree Foundation and Walter and Duncan Gordan Foundation.
- Carter, T. 2015. "Preparing for and Surviving a CRA Audit." Toronto and Ottawa: Carter's Professional Corporation. https://www.carters.ca/pub/seminar/chrchlaw/ott/15/Preparing_Surviving_CRA_audits.pdf
- CCIC (Canadian Council for International Cooperation). 2016a. "Analysis & Results of CCIC/ ICN Survey on Political Activities." Ottawa: CCIC. Unpublished document.
- CCIC. 2016b. Modern Charities, Ancient Rules: Public Policy Activities and Canada's Global Development Sector. Ottawa: CCIC.
- Chan, K. 2016. "The Function (Or Malfunction) of Equity in the Charity Law of Canada's Federal Courts." Canadian Journal of Comparative and Contemporary Law 2 (33): 33–63.
- Chewinski, M. and C. Corrigall-Brown. 2020. "Channeling advocacy? Assessing how funding source shapes the strategies of environmental organizations." Social Movement Studies 19 (2): 222–40.

- Clark, C. 2010. "Canadian Aid Groups Told to Keep Quiet on Policy Issues." Globe and Mail. https://www.theglobeandmail.com/news/politics/canadian-aid-groups-told-to-keep-quiet-on-policy-issues/article1215648/ (February 11).
- Clément, D. 2017. "State Funding for Human Rights Activism: Channeling Protest?" *American Behavioral Scientist* **61** (13): 1703–28.
- Consultation Panel on the Political Activities of Charities. 2017. Report of the Consultation Panel on the Political Activities of Charities. Ottawa: Government of Canada.
- Corrigall-Brown, C. 2022. Keeping the March Alive: How Grassroots Activism Survived Trump's America. New York: New York University Press.
- CRA. 2003. "Political Activities." Policy Statement CPS-022. Ottawa: Canada Revenue Agency. https://www.canada.ca/en/revenue-agency/services/charities-giving/charities/policies-guidance/policy-statement-022-political-activities.html
- CRA. 2013. How to Draft Purposes for Charitable Registration. Guidance CG-019. Ottawa: Canada Revenue Agency. https://www.canada.ca/en/revenue-agency/services/charities-giving/charities/policies-guidance/guidance-019-draft-purposes-charitable-registration.html
- CRA. 2019. Public Policy Dialogue and Development Activities by Charities. Guidance CG-027. Ottawa: Canada Revenue Agency. https://www.canada.ca/en/revenue-agency/services/charities-giving/charities/policies-guidance/public-policy-dialogue-development-activities.html
- CRA. n.d. "Advantages and obligations of becoming a registered charity." Ottawa: Canada Revenue Agency. https://www.canada.ca/en/revenue-agency/services/charities-giving/charities/registering-charitable-qualified-donee-status/apply-become-registered-charity/deciding/advantages-obligations-becoming-registered-charity. html
- Department of National Revenue. 1978. Information Circular #78-3: Registered Charities: Political Objects and Activities. Ottawa: Government of Canada.
- DeSantis, G. and N. Mulé, eds. 2017. "Advocacy: A Contested yet Enduring Concept in the Canadian Landscape" in *The Shifting Terrain: Non-Profit Advocacy in Canada*. Montreal and Kingston: McGill-Queen's University Press, 3–32.
- Devlin, R. 2017. "Charities and Political Activities (A Tempest in a Teapot)." *Canadian Tax Journal* **65** (2): 367–78.
- Drache, A. 2002. "Hostage to History: The Canadian Struggle to Modernise the Meaning of Charity." *Third Sector Review* 8 (1): 39–65.
- Elson, P. 2011. High Ideals and Noble Intentions: Voluntary Sector-Government Relations in Canada. Toronto: University of Toronto Press.
- Elson, P. and S. Hall. 2016. "System Change Agents: A Profile of Public Policy-focused Grantmaking Foundations." Working Paper #9. Montreal: Research Laboratory on Canadian Philanthropy.
- Flyvbjerg, B. 2011. "Case study." *The Sage handbook of qualitative research*. Thousand Oaks, CA: Sage, 301–16.
- Franks, C. 1971. "The dilemma of the standing committees of the Canadian House of Commons." Canadian Journal of Political Science 4 (4): 461–76.
- Gerring, J. 2008. "Case selection for case-study analysis: Qualitative and quantitative techniques." In *Oxford Handbook of Political Methodology*, eds. J. Box-Steffensmeier, H. Brady, and D. Collier. Oxford: Oxford University Press, 645–84.
- Gibbins, R. 2016. "The Moral Imperative for Policy Advocacy." *The Philanthropist Journal*. https://thephilanthropist.ca/2016/02/the-moral-imperative-for-policy-advocacy/
- Grant, A. 2015. Political Activities and the Meaning of "Substantially All"—An Analysis for Registered Charities. Edmonton: Pemsel Case Foundation.
- Hale, G. 2017. "Charities and Politics—A Dubious Mix." Canadian Tax Journal 65 (2): 379-90.
- Hay, C. 2017. "The Interdependence of Intra-and Inter-subjectivity in Constructivist Institutionalism." Critical Review 29 (2): 235–47.
- Imagine Canada. n.d. "Implications of Becoming a Charity." https://sectorsource.ca/managing-organization/starting-organization/implications-becoming-charity
- International Civil Liberties Monitoring Group. 2021. The CRA's Prejudiced Audits: Counter-Terrorism and the Targeting of Muslim Charities in Canada. Ottawa: ICLMG. https://iclmg.ca/prejudiced-audits/
- Jenkins, J. 1987. "Nonprofit Organizations and Political Advocacy" in *The Nonprofit Sector: A Research Handbook*, ed. Walter W. Powell. New Haven, CT: Yale University Press, 307–31.

- Juneau, C. 2016. "The Canadian Income Tax Act and the Concepts of Charitable Purposes and Activities."
 Edmonton: The Pemsel Case Foundation.
- Kelleher, C. and S. Yackee. 2009. "A Political Consequence of Contracting: Organized Interests and State Agency Decision Making." *Journal of Public Administration Research and Theory* **19** (3): 579–602.
- Kirkby, G. 2014. "An Uncharitable Chill: A Critical Exploration of How Changes in Federal Policy and Political Climate Are Affecting Advocacy-oriented Charities." Victoria: Royal Roads University.
- Laforest, R., ed. 2013. "Muddling through Government-Nonprofit Relations in Canada" in *Government-Nonprofit Relations in Times of Recession*. Montreal and Kingston: McGill-Queen's University Press, 9–18.
- Lang, S. 2012. NGOs, civil society, and the public sphere. Cambridge: Cambridge University Press.
- Larsson, O. 2018. "Advancing Post-structural Institutionalism: Discourses, Subjects, Power Asymmetries, and Institutional Change." Critical Review 30 (3-4): 325-46.
- Lasby, B. and E. Cordeaux. 2016. Imagine Canada's Sector Monitor. October. https://sectorsource.ca/sites/default/files/resources/files/sector_monitor_public_policy_activity_2016.pdf
- Lauzière, M. 2021. "A Lever for Change: How Foundations Can Support Public Policy Advocacy" in Intersections and Innovations: Change for Canada's Voluntary and Nonprofit Sector, ed. Susan Phillips. Edmonton: Muttart Foundation.
- Leroux, K. and T. Goerdel. 2009. "Political Advocacy by Nonprofit Organizations: A Strategic Management Explanation." *Public Performance and Management Review* **32** (4): 514–36.
- Levasseur, K. 2012. "In the Name of Charity: Institutional Support for and Resistance to Redefining the Meaning of Charity in Canada." *Canadian Public Administration* **55** (2): 181–202.
- Levi, M. and G. Murphy. 2006. "Coalitions of Contention: The Case of the WTO Protests in Seattle." Political Studies 54 (4): 651–70.
- Liberal Party of Canada. 2015. "Real Change: A New Plan for a Strong Middle Class." https://s3.documentcloud.org/documents/2484248/liberal-party-of-canada-2015-platform.pdf
- Lorinc, J. 2020. "Charitable Sector Gradually Adjusting to New Regulatory World that Allows Unlimited Engagement in Public Policy Dialogue" *The Philanthropist Journal*. March 20.
- Lu, J. 2016. "Fear the Government? A Meta-analysis of the Impact of Government Funding on Nonprofit Advocacy Engagement." *The American Review of Public Administration* **48** (3): 203–18.
- McCarthy, J. and M. Zald. 1977. "Resource Mobilization and Social Movements: A Partial Theory." American Journal of Sociology 82 (6): 1212–41.
- McGregor-Lowndes, M. and B. Wyatt, eds. 2017. Regulating Charities: The Inside Story. London: Routledge. Monohan, P. and E. Roth. 2000. Federal Regulation of Charities: A Critical Assessment of Recent Proposals for Legislative and Regulatory Reform. Toronto: York University Centre for Public Law and Public Policy.
- Mosley, J. 2011. "Institutionalization, Privatization, and Political Opportunity: What Tactical Choices Reveal about the Policy Advocacy of Human Service Nonprofits." *Nonprofit and Voluntary Sector Quarterly* **40** (3): 435–57.
- Mosley, J. 2012. "Keeping the Lights On: How Government Funding Concerns Drive the Advocacy Agendas of Nonprofit Homeless Service Providers." Journal of Public Administration Research and Theory 22 (4): 841–66.
- Moulton, S. and A. Eckerd. 2012. "Preserving the Publicness of the Nonprofit Sector: Resources, Roles, and Public Values." *Nonprofit and Voluntary Sector Quarterly* **41** (4): 656–85.
- National Advisory Council on Voluntary Action. 1977. People in Action: Report of the National Advisory Council on Voluntary Action to the Government of Canada. Ottawa: Secretary of State, Government of Canada.
- Neumayr, M., U. Schneider and M. Meyer. 2013. "Public Funding and Its Impact on Nonprofit Advocacy." Nonprofit and Voluntary Sector Quarterly 44 (2): 297–318.
- Nicholson-Crotty, J. 2007. "Politics, policy, and the motivations for advocacy in nonprofit reproductive health and family planning providers." *Nonprofit and Voluntary Sector Quarterly* **36** (1): 5–21.
- North, D. 1990. Institutions, Institutional Change and Economic Performance. Cambridge: Cambridge University Press.
- Ontario Superior Court of Justice. 2018. Canada Without Poverty v. Attorney General of Canada. 142 O.R. (3d) 754 2018 ONSC 4147.
- Onyx, J. et al. 2010. "Advocacy with gloves on: The "manners" of strategy used by some third sector organizations undertaking advocacy in NSW and Queensland." VOLUNTAS: International Journal of Voluntary and Nonprofit Organizations 21: 41–61.

- Panel on Accountability and Governance in the Voluntary Sector. 1999. Building on Strength: Improving Governance and Accountability in Canada's Voluntary Sector. https://sectorsource.ca/sites/default/files/resources/files/2458_Book.pdf
- Parachin, A. 2007. "Distinguishing Charity and Politics: The Judicial Thinking behind the Doctrine of Political Purposes." *Alberta Law Review* **45**: 871.
- Parachin, A. 2015. "Charity, Politics and Neutrality." The Charity Law and Practice Review 18: 23-56.
- Parachin, A. 2016. "Reforming the Regulation of Political Advocacy by Charities: From Charity under Siege to Charity under Rescue." *Chicago-Kent Law Review* **91** (3): 1047–78.
- Parachin, A. 2017a. "Policy Forum: How and Why to Legislate the Charity-Politics Distinction Under the Income Tax Act." Canadian Tax Journal 65 (2): 391–418.
- Parachin, A. 2017b. "Shifting Legal Terrain: Legal and Regulatory Restrictions on Political Advocacy by Charities" in *The Shifting Terrain: Non-Profit Advocacy in Canada*, eds. Nick J. Mulé and Gloria DeSantis. Montreal and Kingston: McGill-Queen's University Press.
- Pearson, H. 2022. From Charity to Change: Inside the World of Canadian Foundations. Montreal and Kingston: McGill-Queen's University Press.
- Pekkanen, R., S. Smith and Y. Tsujinaka, eds. 2014. Nonprofts and Advocacy: Engaging Community and Government in an Era of Retrenchment. Baltimore, MD: Johns Hopkins University Press.
- Pemsel Case Foundation. 2021. "The Pemsel Case." https://www.pemselfoundation.org/about-us/the-pemsel-case-foundation/
- Ruffa, C. 2020. "Case study methods: Case selection and case analysis" in The SAGE handbook of research methods in political science and international relations, eds. R. Franzese and L. Curini. Thousand Oaks, CA: Sage, 1133–47.
- Salamon, L. 2002. "Explaining Nonprofit Advocacy: An Exploratory Analysis." Working Paper Series. Johns Hopkins University Center for Civil Society Studies, Baltimore, MD.
- Schmidt, V. 2017. "Theorizing Ideas and Discourse in Political Science: Intersubjectivity, Neo-Institutionalisms, and the Power of Ideas." Critical Review 29 (2): 248–63.
- Seawright, J. and J. Gerring. 2008. "Case selection techniques in case study research: A menu of qualitative and quantitative options." *Political research quarterly* **61** (2): 294–308.
- Senate of Canada. 2019. Catalyst for Change: A Roadmap to a Stronger Charitable Sector: Report of the Senate Special Committee on the Charitable Sector. Ottawa: Parliament of Canada.
- Singer, S. 2020. "Charity Law Reform in Canada: Moving from Patchwork to Substantive Reform." *Alberta Law Review* 57 (3): 683–714.
- Skogstad, G. 1985. "Interest Groups, Representation and Conflict Management in the Standing Committees of the House of Commons." *Canadian Journal of Political Science* **18** (4): 739–72.
- Smillie, I. 2012. "Tying up the Cow: CIDA, Advocacy, and Public Engagement." In Struggling for Effectiveness: CIDA and Canadian Foreign Aid, ed. S. Brown. Montreal and Kingston: McGill-Queen's University Press, 269–86.
- Staggenbord, S. 1986. "Coalition Work within the Pro-choice Movement: Organizational and Environmental Opportunities and Obstacles." Social Problems 33 (5): 375–90.
- Sussman, A. 2007. The Art of the Possible: A Handbook for Political Activism. Toronto: McClelland & Stewart
- Trudeau, J. 2015. Mandate Letter to the Minister of National Revenue. Ottawa: Prime Minister's Office. https://pm.gc.ca/en/mandate-letters/2015/11/12/archived-minister-national-revenue-mandate-letter
- Voluntary Sector Initiative. 2002. Advocacy: The Sound of Citizens' Voices: A Position Paper From the Advocacy Working Group. Ottawa: Voluntary Sector Secretariat.
- Watson, R. 1985. "Charity and the Canadian Income Tax Act: An Erratic History." *The Philanthropist* 5 (1): 3–21.

Cite this article: Cameron, John D. and Lauchlan T. Munro. 2025. "A Thaw in the Chill? Charities and Public Policy Engagement in Canada." Canadian Journal of Political Science 58 (1): 95–125. https://doi.org/10.1017/S0008423924000623

	Revenue (2022)*	# Staff (FT + PT) (2022)*	Primary focus (service provision or advocacy)	Federal government funding as % of total revenue (2022)	# Years Political Activities Reported to CRA, 2000-2018	Total # Reports to Lobby Commissioner (2008-2022)	# Lobby Reports on subject of International Development and related subjects (2008- 2022)	FAAE Standing Committee (Briefs + Testimonies), 2000-2022	FINA Standing Committee (Briefs + Testimonies), 2000-2022
Action Canada for Sexual Health and Rights / Action pour la santé et les droits sexuels	\$ 5,643,819.00	34	Service Provision	53.2%	9	1	1	1	14
Aga Khan Foundation Canada	\$ 98,997,573.00	61	Service Provision	25.4%	0	226	185	5	0
Amnesty International - English section	\$17,202,667.00	48	Advocacy	0.0%	14	0	0	38	0
Avocats Sans Frontiers / Lawyers Without Borders	\$ 14,739,356.00	63	Service Provision	92.4%	0	1	1	0	0
British Columbia Council for International Cooperation	\$ 726,646.00	17	Service Provision	70.7%	3	0	0	0	4
Canada World Youth	\$1,369,063.00	49	Service Provision	81.1%	0	39	23	0	0
Canadian Association for the Study of International Development (CASID)	\$ 64,528.00	0	Service Provision	30.6%	0	0	0	0	0
Canadian Catholic Organization for Development and Peace	\$ 25,695,852.00	58	Service Provision	45.8%	15	0	0	7	0
Kentro Christian Network (formerly Canadian Christian Relief and Development Association)	\$203,754.00	4	Service Provision	2.2%	0	0	0	0	0
Canadian Council of Churches	\$1,228,533.00	29	Service Provision	5.1%	4	0	0	0	0
Canadian Crossroads International	\$ 11,387,769.00	52	Service Provision	61.8%	0	0	0	0	0
Canadian Feed the Children	\$ 8,047,194.00	48	Service Provision	6.2%	0	0	0	3	0
Canadian Foodgrains Bank	\$ 63,971,646.00	56	Service Provision	51.5%	18	359	354	8	1
Canadian Friends Service Committee	\$ 1,105,632.00	5	Service Provision	6.1%	18	0	0	0	0

	Revenue (2022)*	# Staff (FT + PT) (2022)*	Primary focus (service provision or advocacy)	Federal government funding as % of total revenue (2022)	# Years Political Activities Reported to CRA, 2000-2018	Total # Reports to Lobby Commissioner (2008-2022)	# Lobby Reports on subject of International Development and related subjects (2008- 2022)	FAAE Standing Committee (Briefs + Testimonies), 2000-2022	FINA Standing Committee (Briefs + Testimonies), 2000-2022
Canadian Lutheran World Relief	\$ 11,882,706.00	24	Service Provision	50.0%	0	10	10	0	0
Canadian Red Cross	\$ 837,143,256.00	7,050	Service Provision	58.0%	3	368	82	13	0
Canadian Association for Global Health (formerly Canadian Society for International Health)		13	Service Provision	58.8%	0	0	0	3	0
CARE Canada	\$ 57,270,529.00	90	Service Provision	87.9%	4	249	211	16	1
Children Believe (formerly Christian Children's Fund of Canada)	\$ 29,301,011.00	127	Service Provision	10.6%	0	0	0	1	0
CODE	\$ 9,331,887.00	36	Service Provision	55.3%	0	0	0	0	0
CoDevelopment Canada	\$ 1,026,409.00	8	Service Provision	0.0%	10	0	0	0	0
Collaboration Sante Internationale	\$ 968,733.00	12	Service Provision	0.0%	0	0	0	0	0
Cooperation Canada (formerly Canadian Council for International Cooperation - CCIC)	\$ 1,984,871.00	21	Advocacy	32.8%	19	617	580	25	6
Council of Canadians with Disabilities	\$299,692.00	5	Advocacy	78.8%	0	0	0	0	0
CUSO International	\$ 31,692,972.00	126	Service Provision	35.3%	0	0	0	3	0
Equality Fund (formerly MATCH International Women's Fund)	\$ 21,755,195.00	58	Service Provision	27.3%	0	112	112	2	0
Equitas - International Centre for Human Rights Education	\$ 6,573,915.00	71	Service Provision	70.0%	0	0	0	0	0
Farm Radio International	\$ 10,820,922.00	30	Service Provision	10.0%	0	0	0	0	0
Forum of Federations	\$ 7,324,214.00	17	Service Provision	80.9%	0	0	0	0	0
Hunger Project	\$ 207,818.00	3	Service Provision	0.0%	0	0	0	0	0
Inter Pares	\$ 7,911,262.00	24	Service Provision	75.8%	7	165	89	7	0
									(Continued)

(Continued)

122

John D. Cameron and Lauchlan T. Munro

Canadian Journal of Political Science

123

	Revenue (2022)*	# Staff (FT + PT) (2022)*	Primary focus (service provision or advocacy)	Federal government funding as % of total revenue (2022)	# Years Political Activities Reported to CRA, 2000-2018	Total # Reports to Lobby Commissioner (2008-2022)	# Lobby Reports on subject of International Development and related subjects (2008- 2022)	FAAE Standing Committee (Briefs + Testimonies), 2000-2022	FINA Standing Committee (Briefs + Testimonies), 2000-2022
Interagency Coalition on AIDS and Development	\$1,334,047.00	5	Advocacy	86.4%	1	0	0	0	0
Islamic Relief	\$ 81,308,685.00	137	Service Provision	2.5%	0	22	22	2	0
Kairos- Canadian Ecumenical Justice Initiatives (Operated under the charity registration of the United Church of Canada until 2023)	\$ 6,785,368.00	15	Service Provision	n/a	Filed under United Church	Filed under United Church	Filed under United Church	14	13
Mennonite Central Committee Canada	\$ 50,831,358.00	120	Service Provision	13.5%	9	143	86	3	2
Mission Inclusion (formerly L'Oeuvre Leger)	\$ 3,023,771.00	n/a	Service Provision	0.0%	0	0	0	0	0
Oxfam - Canada	\$ 34,158,255.00	97	Service Provision	53.3%	13	165	77	12	11
Oxfam - Quebec	\$ 35,073,555.00	97	Service Provision	77.4%	4	194	121	2	2
Pacific People's Partnership Association	\$ 229,936.00	13	Service Provision	23.9%	0	0	0	0	0
Plan International Canada	\$ 276,754,421.00	326	Service Provision	18.3%	1	246	233	0	1
Presbyterian World Service and Development (Operates under charity registration of the Presbyterian Church of Canada)	\$ 4,073,313.00	n/a	Service Provision	n/a	0	0	0	0	0
Primates World Relief and Development Fund	\$ 10,966,190.00	27	Service Provision	26.9%	0	0	0	1	0
Right to Play	\$ 47,888,746.00	521	Service Provision	27.4%	0	97	43	3	0
Save the Children Canada	\$ 69,232,357.00	94	Service Provision	70.0%	5	16	16	14	0
SeedChange (formerly USC Canada)	\$ 6,724,067.00	36	Service Provision	33.8%	7	9	9	0	0

(Continued)

John D. Cameron and Lauchlan T. Munro

	Revenue (2022)*	# Staff (FT + PT) (2022)*	Primary focus (service provision or advocacy)	Federal government funding as % of total revenue (2022)	# Years Political Activities Reported to CRA, 2000-2018	Total # Reports to Lobby Commissioner (2008-2022)	# Lobby Reports on subject of International Development and related subjects (2008- 2022)	FAAE Standing Committee (Briefs + Testimonies), 2000-2022	FINA Standing Committee (Briefs + Testimonies), 2000-2022
SOCODEVI (Fondation SOCODEVI Pour le Developpment International)	\$ 97,317.00	n/a	Service Provision	0.0%	0	0	0	0	0
SOS Children's Villages Canada / SOS Villages d'enfants Canada	\$ 13,391,317.00	22	Service Provision	23.9%	0	6	6	2	0
Steelworkers Humanity Fund (Includes Canadian Network on Corporate Accountability)	\$ 3,279,481.00	8	Service Provision	0.0%	16	203	106	0	0
United Church of Canada (See also Kairos, which operated under the CRA charity registration of the United Church until 2022)	\$ 50,222,000.00	177	Service Provision	6.9%	17	186	51	1	0
Veterinarians Without Borders	\$ 1,958,378.00	7	Service Provision	84.6%	0	0	0	1	0
Victoria International Development Education Association (VIDEA)	\$ 2,766,036.00	84	Service Provision	83.1%	0	0	0	0	0
WaterAid (formerly WaterCan)	\$ 3,605,660.00	11	Service Provision	51.9%	3	0	0	0	1
World Literacy of Canada	n/a	n/a	Service Provision	n/a	0	0	0	0	0
World Renew	\$ 29,502,592.00	42	Service Provision	0.0%	0	2	1	0	0
World Animal Protection (formerly World Society for the Protection of Animals)	\$ 6,307,224.00	27	Service Provision	0.0%	15	209	22	1	0
World University Service of Canada (WUSC)	\$ 40,023,460.00	124	Service Provision	71.4%	0	28	28	7	1
World Vision Canada	\$ 438,744,594.00	394	Service Provision	8.5%	19	647	647	18	9 (Continued)

(Continued.) # Lobby Reports on subject of # Years Federal government Political International FAAE Standing FINA Standing funding as Total # Reports Development Activities Committee Committee Primary focus % of total and related (Briefs + (Briefs + Reported to Lobby Testimonies), # Staff (FT + (service provision to CRA, subjects (2008-Testimonies), Commissioner revenue Revenue (2022)* PT) (2022)* or advocacy) (2022)2000-2018 (2008-2022) 2022) 2000-2022 2000-2022 YMCA Canada (The National \$ 16,533,788.00 Service Provision 25.0% 0 331 10 2 0 19 Council of Young Men's Christian Associations of Canada)

^{*}Data on 2022 revenue and staff numbers is from the CRA charity listings web page: https://apps.cra-arc.gc.ca/ebci/hacc/srch/pub/dsplyBscSrch?request_locale=en